Author's response to reviews

Title: The burden of varicella from a parent's perspective and its societal impact in The Netherlands: an Internet survey

Authors:

Judith H Wolleswinkel-van den Bosch (wolleswinkel@pallas-healthresearch.com)
Anouk M Speets (speets@pallas-healthresearch.com)
Hans C Rümke (rumke@vaxinostics.com)
Pearl D Gumbs (pearl.d.gumbs@gsk.com)
Sander C Fortanier (sander.c.fortanier@gsk.com)

Version: 4 Date: 4 November 2011

Author's response to reviews:

Dear Editor,

In response to your comments of October 26, I have made the following changes to the document:

1. I have made the title identical to that in the submission system, which meant changing capitalisation at some points.

2. With regard to referencing table 2 first. The first reference to a table (table 2) was in the Methods section. I have removed that reference, and only refer to tables in the Results section. By doing so, Table 1 is referenced first.

3. I have changed the email addresses format as you requested.

4. I was not sure what you meant by 'coloured elements' in the tables. Could you please make this change for me if it is still not according to your standard.

Best regards,
Judith Wolleswinkel